Author’s response to reviews

Title: Psychometric properties of the IDS-SR30 for the assessment of depressive symptoms in Spanish population

Authors:

Margalida Gili (mgili@uib.es)
Juan V Luciano (jvluciano@pssjd.org)
Natalia Bauzá (nataliasiddons@hotmail.com)
Jaume Aguado (jaguado@pssjd.org)
Maria J Serrano (mj.serrano@uib.es)
Silvia Armengol (silvia.armengol@almirall.com)
Miquel Roca (mroca@uib.es)

Version: 4 Date: 14 September 2011

Author’s response to reviews:

Changes:
1. The list of authors in the manuscript have been written exactly as they are in the submission system.
2. All the affiliations have been written in the correct format.
3. Reference 47 has been removed from the references because it is not cited in the main manuscript. Therefore, the numbers of the last two references have been modified.
4. We’ve removed the phrase highlighted “the Clinical Research Ethics Committee of the Teknon Foundation”